Author's response to reviews

Title: Curcumin activates the p38MAPK-HSP25 pathway in vitro but fails to attenuate diabetic nephropathy in DBA2J mice despite urinary clearance documented by HPLC

Authors:

Jun Ma (jake.jun.ma@gmail.com)
Lynetta Phillips (lynettaphillips@gmail.com)
Ying Wang (ywang@labiomed.org)
Tiane Dai (tianedai@hotmail.com)
Janine LaPage (jlapage@labiomed.org)
Rama Natarajan (rnatarajan@coh.org)
Sharon G Adler (sadler@labiomed.org)

Version: 6 Date: 10 November 2010

Author's response to reviews:

We have fixed the references using consecutive numbers in square brackets.

Thanks,

Sharon Adler, MD